Author's response to reviews

Title: How well do blood folate concentrations predict dietary folate intakes in a sample of Canadian lactating women exposed to high levels of folate? An observational study

Authors:
Lisa A Houghton (lisa.houghton@acadiau.ca)
Kelly L Sherwood (kellysherwood@yahoo.ca)
Deborah L O'Connor (deborah_l.oconnor@sickkids.ca)

Version: 6  Date: 19 October 2007

Author's response to reviews:

Dear BioMed Central Editorial Production Team:

We have made the formatting changes to the manuscript suggested in your e-mail dated October 11th, 2007. We trust the manuscript is now ready for publication.

Sincerely, Deborah L O'Connor